

# IFPRI Discussion Paper 01323

# February 2014

# Land, Assets, and Livelihoods

Gendered Analysis of Evidence from Odisha State in India

**Vivien Savath** 

**Diana Fletschner** 

**Amber Peterman** 

**Florence Santos** 

Poverty, Health, and Nutrition Division

#### INTERNATIONAL FOOD POLICY RESEARCH INSTITUTE

The International Food Policy Research Institute (IFPRI), established in 1975, provides evidence-based policy solutions to sustainably end hunger and malnutrition and reduce poverty. The Institute conducts research, communicates results, optimizes partnerships, and builds capacity to ensure sustainable food production, promote healthy food systems, improve markets and trade, transform agriculture, build resilience, and strengthen institutions and governance. Gender is considered in all of the Institute's work. IFPRI collaborates with partners around the world, including development implementers, public institutions, the private sector, and farmers' organizations, to ensure that local, national, regional, and global food policies are based on evidence. IFPRI is a member of the CGIAR Consortium.

#### **AUTHORS**

Vivien Savath (viviens@landesa.org) is a research and evaluation specialist at Landesa, Seattle, WA

**Diana Fletschner** is the senior gender expert and director of research at Landesa, Seattle, WA.

**Amber Peterman** was a research fellow in the Poverty, Health, and Nutrition Division of the International Food Policy Research Institute, Washington, DC, when this research was conducted. She is currently an assistant research professor at the University of North Carolina at Chapel Hill, based in Seattle, WA.

Florence Santos is a senior research and evaluation specialist at Landesa, Seattle, WA

#### **Notices**

IFPRI Discussion Papers contain preliminary material and research results and are circulated in order to stimulate discussion and critical comment. They have not been subject to a formal external review via IFPRI's Publications Review Committee. Any opinions expressed are those of the author(s) and do not necessarily reflect the policies or opinions of IFPRI.

Copyright 2014 International Food Policy Research Institute. All rights reserved. Sections of this material may be reproduced for personal and not-for-profit use without the express written permission of but with acknowledgment to IFPRI. To reproduce the material contained herein for profit or commercial use requires express written permission. To obtain permission, contact the Communications Division at ifpri-copyright@cgiar.org.

# **Contents**

Abstract	v
Acknowledgments	vi
1. Introduction	1
2. Background	2
3. Methods	6
4. Results	10
5. Discussion	19
Appendix: Supplementary Tables	21
References	25

# **Tables**

4.1 Preliminary association between households' landownership and their food security	10
4.2 Distribution of households by livelihood activity	11
4.3 Profile of households by livelihood strategy	12
4.4 Basic Model: Multinomial logit model to determine households' livelihood strategies (N = 1,695)	14
4.5 Gendered Model: Multinomial logit model to determine households' livelihood strategies $(N = 1,578)$	16
A.1 Sample composition ( $N = 1,730$ )	21
A.2 Descriptive summary of variables used in the Basic Model $(N = 1,695)$	21
A.3 Descriptive summary of variables in the Gendered Model ( $N = 1,578$ )	22
A.4 Summary of significant results using alternative estimations of basic land documentation model	23
A.5 Summary of significant results using alternative estimations of gendered land documentation model	24

#### **ABSTRACT**

Although asset-based approaches for studying poverty have shown that the portfolio of assets households own or can access influences livelihood strategies and a variety of development outcomes, there is little research unpacking gendered dimensions of asset ownership in diverse contexts. Using data collected from the evaluation of two government land titling interventions in the Indian state of Odisha, this paper examines key relationships linking land and livelihood strategies. The investigation is one of the first to explicitly use the Gender, Agriculture, and Assets Project framework to gain additional insights on how gender—asset dynamics relate to household livelihood strategies. Our results point to a gender-segregated wage labor market, where employment opportunities for rural women are limited; education for both men and women can enable the adoption of more food secure livelihood strategies; and a significant link exists between households that adopt more food secure livelihood strategies and the amount of land they can access, whether they own the land, and the share of land owned by the woman. These results suggest that development interventions to enable households to adopt improved livelihood strategies must consider the gendered context in which they operate, including men and women's employment opportunities, their skills and asset holdings, and make explicit efforts to address constraints in order to facilitate improved development outcomes.

Keywords: land, livelihoods, assets, gender, India

#### **ACKNOWLEDGMENTS**

The authors would like to thank the Gender, Agriculture, and Assets Project (GAAP) team at the International Food Policy Research Institute (IFPRI) and the International Livestock Research Institute (ILRI) for the funding necessary to carry out this research. Agnes Quisumbing and Ruth Meinzen-Dick at IFPRI provided valuable input on data collection tool development and helped launch the fieldwork. The authors would also like to express gratitude to Landesa's program team in Odisha, Sanjoy Patnaik and Sarita Pradhan, for facilitating the fieldwork, and New Concept for conducting the survey work. Ashok Sircar and Santosh Kumar Jha at Landesa's national office provided much appreciated logistics and study design support. Angela Cappoccia and Bijetri Bose, research assistants at Landesa, provided support in the form of desk research and data management.

#### 1. INTRODUCTION

Recent evidence from the empirical study of structural and chronic poverty argues that rather than income or expenditures, it is assets that play a central role in households' ability to exit poverty (Carter and Barrett 2006). This so-called poverty trap literature suggests that effective poverty reduction efforts are contingent on a household's access to a minimum bundle of assets and access to markets to further accumulate assets (Adato, Carter, and May 2006). We argue that who within the household has secure access and control over those assets matters as well. Two government land titling programs operating in the eastern Indian state of Odisha provide an opportunity to examine these asset—poverty hypotheses, focusing on land as a livelihood asset. Accounting for the nuances of land tenure, local context, and gender dynamics, we exploit this opportunity to examine how ownership of land shapes the set of livelihood activities that households can undertake and, as a result, their food security and well-being.

This paper is developed in partnership with IFPRI and the International Livestock Research Institute's Gender, Agriculture, and Assets Project (GAAP). It represents one of the first empirical applications of the GAAP conceptual framework, an extension of the well-known sustainable livelihoods (SL) framework that explicitly considers gender in every aspect and relationship. We also undertake a nuanced examination of asset ownership that considers multiple dimensions of land tenure. Specifically, we examine whether landownership shapes the set of livelihood strategies available to a household; to what extent its influence varies depending on whether a household simply reports owning land or has formal documents to prove its ownership; and how that land was acquired.

The analysis in this paper was carried out in five steps. First, we created an exhaustive typology of livelihood strategies that describes the combination of economic activities in which households in our sample were engaged. We created this typology using narratives from our qualitative research and employing hierarchical cluster analysis to identify patterns in our household survey data. Second, we determined each household's livelihood strategy. Third, we calculated average food security outcomes for households engaged in each livelihood strategy. Fourth, assuming that households put a premium on food security, we ranked the livelihood strategies based on the food security outcomes they were expected to yield. Although conceptually food security represents only one aspect of well-being, the average rural Indian household spends 52.9 percent of its income on food, making it a significant decision factor in livelihood choices (NSSO 2013). Fifth, we relied on econometric analysis to assess the extent to which having land influenced households' choice of livelihood strategy and, consequently, their food security.

In our analysis, we are able to distinguish among land that households access, land that they own, and land for which they have formally recognized documents. We also distinguish land that was inherited from land that was purchased or received from the government. Finally, because gender dynamics may shape these results, we replicate the analysis using sex-disaggregated measures of asset ownership.

The paper is organized as follows. Section 2 provides background for the analysis, describing the analytic framework, the context, and the land policies and government programs in Odisha. Section 3 describes the methodology used in this study and the data collected. Section 4 presents our findings. Section 5 concludes with a discussion of policy implications and directions for future research.

#### 2. BACKGROUND

# Sustainable Livelihoods and Gender, Agriculture, and Assets Project Frameworks

Unlike traditional approaches for studying poverty that emphasize income as the basis for poverty and vulnerability, the SL framework considers the entire portfolio of assets that households own or can access; the impact of owning or having access to these assets on household livelihood strategies; and how household assets and livelihood strategies affect a variety of outcomes, such as household food security and vulnerability to shocks and seasonality. The framework was created in response to the recognition that standard definitions of poverty based on income and consumption applied in developed countries were narrow and inadequate for lower income economies that were structurally very different (Ellis 2000; Bebbington 1999; Moser 1998; Chambers 1995; Chambers and Conway 1991). The SL framework has been widely adopted by those working in rural poverty research and alleviation because it relies on a mixed-methods contextualized approach, because it considers a wider portfolio of strategies to manage and cope with the risks associated with shocks and seasonality, and because of its recognized potential to better guide the design of effective interventions in developing economies (Ashley and Carney 1999a, 1999b; Scoones 1998).

Under the SL framework, a livelihood activity is defined as any direct income-generating activity in which an individual engages (for example, dairy production or agriculture) or any activity that might not directly bring in income but increases the consumption and/or well-being of an individual (for example, firewood collection or cultivation of medicinal herbs). A household's livelihood strategy, in turn, is the combination of livelihood activities in which its members engage. The livelihood strategies that we observe are the result of households' options and preferences. Their options depend, at least partially, on their asset endowment.

Assets in the SL framework are more than inputs into income creation because they impact agency and status: "Assets are not simply *resources* that people *use* in building livelihoods: they are assets that give them the *capability* to be and to act" (Bebbington 1999, 2022; italics original). The SL literature focuses on the accumulation of five types of assets, often called the asset pentagon: (1) human assets (for example, labor hours, education, and health); (2) social assets (for example, trust, family support, and community membership); (3) physical assets (for example, house and access to nearby roads); (4) natural assets (for example, land and forest access); and (5) financial assets (for example, cash reserves and credit availability). Other assets, such as political power and psychological assets, are sometimes included as well, although there is lack of agreement on whether and how best to incorporate them into the framework (Scoones 2009; De Haan and Zoomers 2005).

Empirical research in both developed and developing countries has demonstrated that households do not always pool their resources and that it matters who in the household has access to and control over these resources (Haddad and Hoddinott 1997). Yet, the SL framework is silent on gender, hampering its ability to explain how individual ownership of or access to assets influences individual livelihood activities and household livelihood strategies. Meinzen-Dick and colleagues (2011), as part of GAAP framework, expand the SL construct to consider gender relations and how these relations influence livelihood constraints and opportunities, creating a framework suitable for gendered analysis of asset interventions. In the GAAP framework, individuals' gender and the environment in which they live can affect their access to and control over assets, their livelihood goals, and the livelihood activities they choose. In other words, men and women who have access to and control over the same assets and who live in the same environment can make systematically different consumption, savings, and investment decisions.

#### Land as a Livelihood Asset

The SL framework describes two important types of relationships between assets: (1) sequencing, or the degree to which the acquisition of one asset enables the acquisition of another, and (2) substitution, or the degree to which particular assets can be substituted for others. Land is a particularly important asset for rural livelihoods because of its primacy in asset sequencing. Those who own land may be more inclined to invest in natural capital through activities like soil conservation (Bekele and Mekonnen 2010; Pender and Kerr 1996) may be able to use it as collateral to access financial capital (Petracco and Pender 2009); they may also be more likely to invest in their children's education, thereby enhancing their households' human capital (Katz and Chamorro 2002). Land is a marker of social status in many agrarian societies, including rural India, and landowners may therefore also benefit from greater social capital (World Bank, Food and Agriculture Organization, and International Fund for Agricultural Development 2009).

There is ample evidence for gender-differentiated access to and control over land, underscoring the need to use a gendered lens in an analysis of land and livelihood (Deere and Doss 2006; Deere and Leon 2003; Agarwal 1994). Beyond the benefits outlined above, secure access to and control over land are particularly important for women when their households or communities lack formal sources of social security or access to other social safety nets (Giovarelli 2009).

In particular, studies have shown that improvements in women's land rights can be linked to a number of socially and economically desirable outcomes. Improved women's land rights have been associated with (1) lower reported rates of long-term physical and psychological domestic violence (Gupta 2006; Panda 2006); (2) improved family nutrition (Allendorf 2007; Katz and Chamorro 2002); (3) improved children's educational achievements (Katz and Chamorro 2002); (4) decreased fertility (Field 2003); and (5) increases in women's ability to participate in their households' decisionmaking (Santos, Savath, Fletschner and Peterman 2013; Allendorf 2007).

# **Defining Livelihood Activities and Livelihood Strategies**

Research from a range of disciplines shows that, globally, the rural poor employ a vast range of livelihood strategies to meet basic needs and increase their well-being (Nielsen, Rayamajhi, Uberhuaga, Meilby, and Smith-Hall 2012; Ellis 2000). Despite this, many studies use some measure of income-generating activities to classify livelihoods (Babulo et al. 2008; Abdulai and Crole-Rees 2001; Birch-Thomsen, Frederikson, and Sano 2001). Although easy computationally and intuitive in structure, these income-based classifications often hide great variation in the livelihood activities of households. They also suppress information on activities that do not bring in direct income but are still important to household livelihood and well-being.

Furthermore, importantly for our purposes, these approaches are often unable to distinguish between a household that is diversifying its livelihood strategy—by increasing the number of livelihood activities in which it engages, as it attempts, for example, to cope with a shock or to mitigate risks (Ellis 2000)—and a household that is shifting its livelihood strategies because new strategies are available (for example, as a result of changes in the household's asset portfolio).

Cluster analysis allows us to better differentiate between shifts and diversification. As a more computationally intensive approach, cluster analysis uses distance-minimizing methods to sort households' livelihood activity data into similar groups, creating a set of livelihood strategies. By generating data-driven livelihood strategies, this technique allows us to categorize households (each household fits into one livelihood strategy) and, simultaneously, define characteristics of the livelihood strategies themselves (Iiyama, Kariuki, Kristjanson, Kaitibie, and Maitima 2008; Brown, Stevens, Ouma, Murithi, and Barrett 2006; Jansen, Pender, Damon, Wielemaker, and Schipper 2006; Orr and Mwale 2001).

# **Setting: Odisha Geography and Policy Landscape**

Odisha is a coastal state of approximately 155,707 square kilometers in eastern India, with widely varying physiography. The state is typically divided into four physical zones: (1) northern plateau, (2) central river basin, (3) mountainous Eastern Ghats, and (4) coastal plains. The coastal plains zone has the most advanced agriculture due to increased soil fertility and irrigation potential, while other zones of Odisha are often dependent on rainfed agriculture. Our study samples four districts in Odisha from the coastal plains and the Eastern Ghats.

Odisha has historically been one of the poorest states in India, with high rates of poverty and low rates of economic growth and diversification. According to the 2011 census, 83.3 percent of Odisha's population of nearly 42 million people live in rural areas. Nearly 60 percent of the population depend on agriculture to some degree for their livelihoods. As noted before, much of the state is dependent on rain for watering the crops, leaving much of the population exposed to risk from drought and flood. Odisha has experienced at least one natural disaster in 11 of the past 13 years (State Government of Odisha 2011). In addition, the state has a large proportion of historically disadvantaged groups, including the scheduled tribes and scheduled castes, which comprise 22.1 percent and 16.5 percent of the population, respectively (Census of India 2011). The scheduled tribe and scheduled caste designations form the basis for the distribution of many government entitlement programs at the national and state levels.

The current geography of Odisha comprises four formerly distinct land revenue administration systems originally established during the British colonial period. These administration systems included major differences in revenue assessment1 and record keeping, leading to administration difficulties that persist today. Beginning with Indian independence in 1947, Odisha passed numerous laws aimed at land reform and unifying land administration in the state. Implementation of many national reform initiatives, such as the land ceiling laws, and early state-level efforts were minimally effective in large part due to the diverse land administration legacy and lack of reliable records (Deo 2011; Mearns and Sinha 1999). The early legislation also typically subsumed women's interests under those of her household, usually represented in public affairs by the adult male "head of household."

In line with India's *land to the tiller* movement, the Odisha Land Reforms Act included provisions for adverse possession. The burden of proof, however, lay on the cultivating party, whose work was highly unlikely to be documented, and these rights were rarely exercised. Under the Odisha Prevention of Land Encroachment Act, revenue inspectors were the only parties able to identify and regularize land to eligible beneficiaries; however, their incentives typically lay in the direction of collecting fees for encroachment on government land rather than on initiating burdensome regularization processes (Mearns and Sinha 1999). Later state-level programs would aim to rectify these shortcomings, with emphasis on joint or women's titling and targeting the particular challenges facing the scheduled tribe and scheduled caste communities.

#### Vasundhara and Gramakantha Paramboke Land Titling Programs

This study exploits data collected for an evaluation of two Odisha government land titling programs. One program, Vasundhara, grants title to households considered to be encroaching on government land, while the other, Gramakantha Paramboke (GKP), distributes individual title to households residing on previously communally titled land.

<sup>&</sup>lt;sup>1</sup>Two main systems of revenue assessment have a legacy in Odisha:

The *Zamindari* (landlord) System, in which land is an independent property and the landlord deposits tax revenue at the district treasury. There is no consistent administration below this district level and the *zamindars*, or landlords, organize their own revenue collection system in their estate, typically involving several layers of intermediaries.

The *Ryotwar* (peasant-proprietor) System, in which land belongs to the British Crown and is held in a right of occupancy. Village headmen (a heritable office) physically collect revenue from occupants on a 10 percent commission basis and keep records for their own village.

The Vasundhara program was launched in 2004 in response to a state-initiated survey finding high numbers (249,334 households) of absolute landlessness<sup>2</sup> in Odisha. This program aims to provide a homestead plot of at least 4 decimals (decimal = 1/100 acre) in size per beneficiary to all homestead-less rural households (later modified to allow 10 decimals per household where feasible). The program leverages already existing, but heretofore largely unused, provisions in the Odisha Government Land Settlement Act and the Odisha Prevention of Land Encroachment Act that allowed homestead distribution and settlement on government land. The program prioritized regularization of government land on which the household is already residing, but where available land is scarce, Vasundhara allows for land purchase. To be eligible for a homestead under the program, a household should not own any other homestead plot, should earn fewer than 15,000 Indian rupees (Rs.) per annum (approximately US\$270), and should not possess (via tenancy, borrowing, or other arrangements) more than one standard acre of agricultural land. By granting ownership documents, Vasundhara provides greater security and legal status to households previously considered to be encroaching on government land and subject to government fines. Additionally, households are provided basic agricultural extension services to support backyard cultivation. The program's more recent emphasis on titling land jointly for husband and wife is meant to provide women with an additional layer of legal protection.

*GKP* is a separate state government program that provides individual titles to households that were previously on collectively owned land, under the 2009 Amendment of the Odisha Government Land Settlement Act. The land was communally titled as a vestige of the colonial Madras presidency, during which a survey on agricultural land opted to count entire village sites as one individual plot, excluding residents of these villages from any record of rights to that land. As with Vasundhara, the Land Revenue Department is now responsible for identifying those eligible households through enumeration and thereafter providing them *patta*, or formal government deed for land allocated by the Indian government, issued jointly to dual-headed households.

The programs are implemented on the basis of the land classification of a particular geography, which is itself partially a vestige of the multiple colonial land administration systems. This means that though both programs regularize households' property by granting them formal documents to the land on which they already reside, the Vasundhara and *GKP* programs generally do not overlap geographically. The enumeration process for both of these programs is labor- and time-intensive. The programs partner with nongovernmental organizations and quasi-governmental agencies such as Landesa and the Odisha Tribal Empowerment and Livelihoods Program to speed the identification of eligible families and verification of their land status.

<sup>&</sup>lt;sup>2</sup> Absolutely landless refers to households without agricultural land or homestead and is distinct from landlessness, which is often used to refer to households that have no agricultural land (but could potentially own a homestead).

#### 3. METHODS

#### **Qualitative Data Collection**

Qualitative data collection took place in the post-monsoon season, September and October 2012, one to two years after land program beneficiaries in pilot sites received *patta*. We conducted 15 key informant interviews with implementers of the land allocation and regularization programs to better understand the processes and implementation of the interventions. In addition, we conducted 14 life history interviews to provide insight into livelihood strategies and household and intrahousehold decisionmaking. Finally, seven focus group discussions provided a view of community-level norms surrounding landownership and use, as well as a survey of available livelihoods.

Participants for the qualitative work were purposefully sampled from one district to gain an indepth perspective in a specific locality. We chose to do the qualitative work in Ganjam District, the site of Landesa's first pilot activity, to allow the longest possible time period to detect treatment effects. Key informant interviews were sampled to include perspectives from all levels of implementation, from the village to the district administration. Those interviewed through life history interviews and focus group discussions were purposefully selected to include diverse demographic categories, such as single women and scheduled tribes.

#### **Quantitative Data Collection**

This study leverages data collected for an evaluation of the Vasundhara/GKP programs in the areas where Landesa has partnered with the government. We surveyed 1,730 households across three pilot intervention districts and one control district of Odisha in October and November 2012.

The sample was determined before actual beneficiaries entered the program, and therefore our sample design includes households that we deemed likely to become beneficiaries—because they were in blocks where Landesa assisted with the program—as well as households that we thought were less likely to be selected, either because they were in other blocks or because they were occupying land that was harder to regularize.<sup>3</sup> The survey firm, New Concept, worked with local land administration officials to identify program-eligible households to include in the sample.

We interviewed the adult woman in each household using questions to capture the household's demographics, housing condition, physical infrastructure, access to water and sanitation, adult men and women's participation in household decisionmaking, engagement in self-help groups, income-generating activities, expenditures and debt, membership and participation in local social institutions, food security, cultivation of home gardens, and livestock assets.

#### **Analysis**

We hypothesize that households' livelihood options and consequently their chosen livelihood strategy are influenced by access to and rights over land. That is, the livelihood strategy they choose depends on the size of the plots they can access, how long they have had access to that land, whether they own those plots, whether they have documents to prove their ownership, and how they acquired the land—through inheritance, purchase, or government programs. To test these hypotheses we took the following steps. First, we drew upon qualitative and quantitative information to elicit a comprehensive list of livelihood activities observed in the study region, to determine their prevalence and to probe their relation to contextual factors. The qualitative data were coded and analyzed deductively using Nvivo and following GAAP's conceptual framework.

<sup>&</sup>lt;sup>3</sup> Land that is classified by the Land Revenue Department as "leasable" is the easiest to regularize. Households on "non-leasable" land wishing to obtain formal documentation through the *Vasundhara* program must either go through a process to change the land classification or find available leasable land and be willing to move to it.

Second, as proposed by the SL framework, a study of household behavior must consider that households can and often do engage in more than one livelihood activity—the combination of their activities is their livelihood strategy. Thus, we leveraged our quantitative data and relied on a clustering technique described below to determine data-driven livelihood strategies observed among households in our sample.

Third, because one can uniquely match each household in our sample to a livelihood strategy, we used the food security data we had collected to calculate average food security outcomes per livelihood strategy. Assuming that households value food security, we can predict which livelihood strategies are superior.

Fourth, based on this food security ranking of livelihood strategies, we relied on a multinomial logit analysis to identify whether and to what extent households' choices of livelihood strategies are influenced by the amount of land households can access and the rights they have to that land.4 This econometric approach allows us to identify which factors are statistically associated with households choosing one livelihood strategy over another and enables us to determine whether the land-related factors play a role. Moreover, because we can control for other asset-based factors that may also influence households' choices, this approach permits us to assess the magnitude of the land-related effects.

### Clustering Technique Used to Identify Livelihood Strategies

Following researchers such as Alinovi, D'Errico, Erdgin, and Romano (2010) and Jansen, Pender, Damon, Wielemaker, and Schipper (2006), we use cluster analysis to create a taxonomy of the livelihood strategies observed in the sample. Cluster analysis is a highly flexible and intuitive method for assigning a large number of observations to a smaller number of distinct groups, or clusters. In hierarchical cluster analysis, each of the cases begins as a cluster unto itself. Clusters are then successively merged together based on their similarity using Ward's algorithm, which seeks to minimize the merging cost.<sup>5</sup>

In our case, the analysis started by looking at each household's data to derive their livelihood strategy (the set of activities they chose). That is, we started with 1,730 livelihood strategies or clusters by themselves. Comparing each cluster to all the others, the algorithm determined which of them looked sufficiently *similar* and merged them. The algorithm then compared each cluster to all the other clusters and combined those that were similar into bigger clusters, minimizing the merging cost each time. The resulting clusters are the livelihood strategies observed in the sample. Note that by definition, each household belongs to one and only one cluster, or equivalently, has one and only one livelihood strategy.

$$\Delta(A,B) = \sum \|x_i - x\|^2 - \sum_{i \in A} \|x_i - \bar{a}\|^2 - \sum_{i \in B} \|x_i - \bar{b}\|^2 = \frac{n_a n_b}{n_a + n_b} \|\bar{a} - \bar{b}\|^2$$

<sup>&</sup>lt;sup>4</sup> The multinomial logit model relies on the assumption of independence of irrelevant alternatives, which in our case means it assumes that the probability that a household would choose a livelihood strategy *x* over the base livelihood strategy is independent of what other livelihood strategies are available to the household. Although this can be a fairly restrictive assumption, we relied on two robustness checks to determine that the relative probabilities of participating in the livelihood strategies as defined are adequately independent to use the multinomial logit model. First, we used the Small-Hsiao method to test the independence of irrelevant alternatives assumption statistically and failed to reject it. Second, we repeated the analysis using two alternative estimation strategies that do not impose independence of irrelevant alternatives, namely individual logit models and a multivariate probit model. The results from these alternative specifications largely confirm our findings. (See Tables A.3 and A.4 in the appendix.)

<sup>&</sup>lt;sup>5</sup>The *merging cost* is calculated as the increase in the sum of squared distances after merging clusters A and B,  $\Delta(A, B)$ . The following equation describes the merging cost, where A and B are the potential resulting clusters after a merge, *i* indexes the current clusters, and *n* is the number of observations in the resulting clusters.

To determine which strategies/clusters were similar, we used the Jaccard index. This index is widely used for generating taxa in ecological fields and is especially fitting for matching on asymmetric binary variables, such as our household livelihood activity data. Asymmetric variables are those for which the researcher is interested in matching those who *do* participate in particular activities (positive matches), but cannot draw any conclusions about similarities between those who do *not* participate in that activity (negative matches).

Hierarchical clustering requires determining the desired number of clusters in advance. We used the Caliński-Harabasz index to determine the optimal number of clusters (Caliński and Harabasz 1974). This measure suggested that we should aim for four clusters. In other words, our analysis includes 11 livelihood activities and four livelihood strategies.

## **Econometric Estimation of the Relationship between Assets and Livelihood Strategies**

A household's livelihood strategy is understood to be a polychotomous choice, and we followed the existing literature (Nielsen, Rayamajhi, Uberhuaga, Meilby, and Smith-Hall 2012; Jansen, Pender, Damon, Wielemaker, and Schipper 2006) in estimating a multinomial logit model to identify whether and to what extent households' choices of livelihood strategies are influenced by the amount of land households can access and the rights they have to that land. This econometric approach uses a set of parallel equations to allow us to identify which explanatory factors are statistically associated with households choosing livelihood strategy  $j_i$  over a base livelihood strategy  $j_i$ . Moreover, because we can control for other asset-based factors that may also influence households' choices, this approach allows us to isolate the magnitude of the land-related effects. Specifically, the probability  $p_{ij}$  of household i choosing livelihood strategy j is a function of a vector of case-specific regressor variables  $x_i'$  subject to the constraints that  $0 < p_{ij} < 1$  and  $\sum_{j=1}^4 p_{ij} = 1$  where:

$$p_{ij} = \frac{\exp(x_i'\beta_j)}{\sum_{l=1}^4 \exp(x_i'\beta_l)} j = 1, 2, 3, 4.$$

when we set  $\beta_{j=1}$  to zero, we can interpret the coefficients for the alternate,  $j \neq 1$ , livelihood strategies relative to the base strategy. For ease of interpretation, the results we report in this paper refer to the change in probability that households will choose livelihood strategy j rather than the base strategy, where:

$$\frac{\Pr(y_i = j)}{\Pr(y_i = base\ strategy)} = \exp(x_i'\beta_j), j = 1, 2, 3, 4.$$

We estimate these probabilities using two models: a *Basic Model* and a *Gendered Model*. The Basic Model estimates  $x_i'\beta_j$  as a function of (1) land-related variables (LAND) that include land tenure status, land size, and number of years over which the household has had access to these plots; (2) variables related to nonland assets (NLASSETS) that serve as proxies for the household asset endowment; and (3) other household (HH) and community (CC) characteristics, where:

$$x_i'\beta_j = \beta_{j0} + \beta_{j1} * LAND + \beta_{j2} * NLASSETS + \beta_{j3}HH + \beta_{j4}CC + \varepsilon_{ij}, j = 1, 2, 3, 4.$$

The Gendered Model works with the same variables but further considers the share of land over which the male and female heads of household claim ownership and for some of the nonland assets (such as labor force and education) uses sex-disaggregated variables. We chose to use the share of land owned,

$$J(A,B) = \frac{m_{01} + m_{10}}{m_{01} + m_{10} + m_{11}}, m = 1, ..., 11$$

<sup>&</sup>lt;sup>6</sup> The Jaccard distance is calculated with the following formula, where  $m_{ab}$  is a count of livelihood activities that match a particular combination between strategies A and B. For example,  $m_{0l}$  is the number of activities that strategy A *does not* do (0), but strategy B *does* do (1).

rather than the amount, to capture the man/woman's standing relative to other members of his/her household. Shares were counted toward both the man and the woman if a particular asset was reported to be jointly owned, and counted toward neither if someone other than the male and female heads of household was reported as the owner. Thus,

$$\begin{aligned} x_i'\beta_j &= \beta_{j0} + \beta_{j1} * LAND + \beta_{j2}LAND * LANDSHR_f + \beta_{j3}LAND * LANDSHR_m \\ &+ \beta_{j4}NLASSETS + \beta_{j5}NLASSETS_f + \beta_{j6}NLASSETS_m + \beta_{j7}HH + \beta_{j8}CC + \varepsilon_{ij}, j = 1, 2, 3, 4. \end{aligned}$$

Descriptive statistics for all the explanatory variables are presented in Tables A.3 and A.4 in the appendix. The Gendered Model was estimated using a sample limited to households that had both a male and a female head of household.<sup>7</sup>

# **Food Security Variables**

We examined three household-level food security outcomes to rank livelihood strategies. To capture households' dietary diversity we used the Household Dietary Diversity Score, an indicator that summarizes the total number of food groups (out of a possible 12) consumed by the household members in the past 24 hours (Swindale and Bilinsky 2006). As a proxy for a household's vulnerability to hunger we used a binary indicator for whether the household reported experiencing times when it did not have food or money to buy food in the past three months. Finally, we used total household expenditure on food.

<sup>&</sup>lt;sup>7</sup> Because the Gendered Model was likely to be affected by multicollinearity, we tested the independent variables' variance inflation factors (VIF). The resulting mean VIF for all of these models was below 3, signaling a generally acceptable level of multicollinearity (a common, though arbitrary, cut-off is 4).

#### 4. RESULTS

#### **Land Tenure**

Our household survey revealed that all of the households interviewed had some access to land. However, the data we gathered also show that tenure security varies considerably: more than one-third of the households do not own land (38 percent) and another third of them (31 percent) reported owning land but not having formal documents to prove their ownership. The remaining 31 percent of households have documents to prove that they own at least one of the plots they are currently accessing. Among households that own land, the vast majority acquired it through inheritance, typically from the male side.<sup>8</sup>

Table 4.1 Preliminary association between households' landownership and their food security

	Households that	Households that	
Food security indicator	do not own land	own land	
Percentage of sample (N = 1,730)	37.5	62.5	
Household dietary diversity score (mean)	5.51	5.66	
Households vulnerable to hunger (%)	7.4	4.0	
Food expenditure (Rupees/month)	1,375	1,610	

Source: Authors' calculations based on household survey data.

Note: \*\*\* indicates that the mean outcomes are significantly different at the 1% level.

A quick initial probe into the relationship between landownership and food security, based on a simple comparison of means, shows that even though all households have access to some land, those that own land are more food secure: they spend more on food, they are less vulnerable to hunger, and they have a more diverse diet. A further check on how these results vary depending on whether the household has land documents or how their land was acquired shows that although the results remain generally positive, the patterns become less obvious and uncovering them requires a more sophisticated analysis. We devote the remainder of the paper to a robust examination of these relationships between land and livelihoods.

#### Livelihood Activities

Data from the quantitative survey allowed us to determine the most common livelihood activities in the study region. They include cultivation of (1) rice, (2) fruits, and (3) vegetables; rearing of (4) draft animals, (5) cows, (6) small animals, and (7) poultry; working (8) in agriculture, (9) as a nonagricultural wage laborer, or (10) as self-employed; and (11) receiving food or other public transfers. Information about activities 1 through 7 came from households' reported engagement in agriculture and animal husbandry. We classified them into these seven categories recognizing that these activities differ considerably in the investments they tend to require, the length of time they take to produce results, and the level of revenues they typically yield. Information about activities 8 through 10 was captured when we inquired about the main occupation (whether or not it was compensated) of each household member over the age of seven during the past three months. Unfortunately, because our questionnaire asked only about their main activity, it is possible that we missed activities that, while important, were not a household member's main activity.

<sup>&</sup>lt;sup>8</sup> For households that own more than one plot of land and whose plots were acquired through multiple sources—8.73 percent of the sample—we classified them as acquiring land from the source from which they had obtained their largest plot.

Lastly, we identified households that participated in government entitlement and support programs. These mainly consisted of the Public Distribution System (PDS), a national program for distributing price-controlled foods through ration shops. The PDS is notoriously rife with inclusion and exclusion errors, and Landesa's experience over the years indicates that accessing these programs is often incumbent on the recipient and represents a strategic choice much like any other livelihood activity.

Table 4.2 provides a distribution of households by livelihood activity. As shown, nearly half of the households surveyed reported engaging in agricultural wage labor. This might be partially due to the surveyed reference period of the past three months, which happened to coincide with the planting season (October 2012).

"Yes, now weeding is over, then we will go to the forest. Until harvest time we are busy in collecting forest products. Then it will be harvest time." —Life history interview with female beneficiary (SC, age 30)

Table 4.2 Distribution of households by livelihood activity

Livelihood activity	Percentage of households that engaged in each livelihood activity
Livelihood activity  Cultivation	(N = 1,730)
Rice	34.5
Fruits (for example, mango, banana, coconut, papaya)	26.0
Vegetables (for example, green leafy vegetables, pumpkin, beans)	8.2
Livestock Rearing	
Draft animals (horses, oxen)	18.2
Cows	20.9
Small animals (pigs, goats, sheep)	12.7
Poultry (ducks, pigeons, chickens)	9.9
Employment	
Agriculture labor	49.7
Nonagriculture wage labor	32.0
Self-employed	17.2
Government Transfers	
Ration card and other public transfers	64.1

Source: Authors' calculations based on household survey data.

Rice is the most common crop, cultivated by more than a third of the households. Approximately 20 percent of the households engaged in large livestock rearing, both draft animals (most commonly buffalo) and cows. Small livestock and poultry rearing were less common, though the qualitative data suggest that livestock ownership is a common aspiration for which many households are saving money.

Close to one-third of the households engage in nonagricultural wage labor, which consists mainly of earthwork, factory work, and construction. Self-employment, practiced in 17 percent of households, consists of skilled trades such as carpentry and masonry, and small enterprises such as vending.

Finally, 64 percent of the sampled households are receiving public transfers. Most of these households receive subsidized food staples purchased at specific distribution centers of the PDS. In the life history interviews, respondents frequently brought up access to this program as a reason for improvements in their lives.

### **Livelihood Strategies**

The four livelihood strategies we identified by aggregating the 11 livelihood activities described above using the hierarchical cluster analysis are:

- 1. Agricultural wage laborer: These are households that earn their living through agricultural work on others' land. They do not engage in their own cultivation or livestock rearing.
- 2. Off-farm wage laborer: These are households whose only source of income is wage labor that is not directly related to agriculture. Most of them do factory or construction work. They do not engage in their own cultivation or livestock rearing.
- 3. Wage laborer with a farm: These are households whose livelihoods tend to combine agricultural and livestock work on their farm with work that they do for others.
- 4. Farm and self-employment: These are households that, in addition to working on their own farm and for others, are likely to have their own income-generating activities.

Table 4.3 presents a profile of the average household for each of the livelihood strategies.

Table 4.3 Profile of households by livelihood strategy

	Livelihood Strategies								
Livelihood activities and food security outcomes	Agricultural wage laborer households	Off-farm wage laborer households	Wage laborer with a farm households	Farm and self- employment households					
Number of households	245	176	995	312					
Livelihood activities in which they engage (%)									
Cultivation									
Rice			55.1	15.0					
Fruits			37.8	23.6					
Vegetables			13.5	2.6					
Livestock rearing									
Draft animals (horses, oxen)			30.2	4.5					
Cows			34.3	6.4					
Small animals (goats, pigs, sheep)			21.6	1.6					
Poultry (ducks, chickens)			17.1	0.3					
Employment									
Agricultural labor	100.0		55.4	19.8					
Other wage labor		100.0	33.3	14.4					
Self-employed			7.0	72.8					
Government transfers									
Ration card and other public transfers Food security outcomes	50.6	52.3	69.9	62.9					
Household dietary diversity score (mean)	5.4	5.6	5.6	5.9 ***					
Percentage of households vulnerable to hunger	15.5	4.6 ***	3.3 ***	4.2 ***					
Food expenditure (Rupees/month)	1,377.3	1,784.0 **	1,439.9	1,749.8 **					

Source: Authors' calculations based on household survey data.

Fourteen percent of the households we surveyed fall into the agricultural wage laborer livelihood strategy. Their sole source of livelihood is agricultural wage work, which yielded a three-month average income of Rs. 4,651 (US\$77) for men and Rs. 2,895 (US\$48) for women. As a group, households following this livelihood strategy exhibit the worst food security indicators: lowest expenditures on food, highest vulnerability to hunger, and a less diversified diet. As such we assume that, given the option, households would prefer any of the other livelihood strategies; we have therefore chosen to use the agricultural wage laborer category as the base livelihood strategy for the econometric analysis that follows.

Off-farm wage laborer households account for approximately 10 percent of the sample. Compared to households following the base livelihood strategy, these households tend to spend more on food and are less vulnerable to hunger, but their diet is not more diversified.

Wage laborer with a farm households make up 58 percent of the sample. Only 3 percent of these households experienced hunger in the previous three months, and their diet is significantly more diverse than those following either of the first two livelihood strategies. We are unable to distinguish to what extent the superior food security outcomes are a result of their productive activities or their ability to join the PDS and receive public transfers—70 percent of the households in this group have received food or cash transfers.

Finally, 18 percent of the households we interviewed follow the fourth livelihood strategy: farm and self-employment. Approximately 73 percent of these households have a member who is self-employed. These households experience relatively low vulnerability to hunger. They are among the ones who spend the most on food and, as a group, they exhibit the most diverse diet. A high proportion of them receive subsidized food or transfers.

# Results from the Basic Model: Assets and Livelihood Strategies

The results presented in Table 4.4 enable us to distinguish which factors, and particularly which assets, may affect households' ability to embrace a given livelihood strategy rather than choosing to be an agricultural wage laborer household. As discussed earlier, in analyzing households' livelihood strategy choices we considered four types of factors: land-related characteristics, households' nonland asset endowments, other household characteristics, and community-level conditions. Because of our focus on land, we tried three versions of this model. We started by only considering households' access to land, how long they had had access to those plots, and whether they owned at least one plot. We then further disaggregated the analysis by paying attention to whether the households had documents to prove their ownership rights. Lastly, we looked at whether the way they had obtained the plot mattered.

Table 4.4 Basic Model: Multinomial logit model to determine households' livelihood strategies (N = 1,695)

Covariates	Probability that households choose the off-farm wage laborer strategy over being an agricultural wage laborer household			choose th	tegy over be ral wage lal	orer with a eing an	Probability that households choose the farm and self-employment strategy over being an agricultural wage laborer household			
	(1)	(2)	(3)	(1)	(2)	(3)	(1)	(2)	(3)	
Baseline probability that households would choose livelihood strategy [] over being an agricultural wage laborer household	10.449**	10.499**	10.480**	0.837	0.834	1.007	0.914	0.922	0.867	
LAND										
Log (land household can access) Log (average number of months since land was	1.116	1.110	1.116	2.161***	2.150***	2.146***	1.408***	1.400***	1.412***	
acquired) Owns land	0.694*** 0.746	0.699***	0.697***	0.912 1.420**	0.923	0.896	0.926 1.357	0.930	0.935	
Owns land, undocumented		0.656			1.161			1.233		
Owns formally documented land Owns inherited land Owns purchased land		0.882	0.744 0.758		1.797***	1.593** 1.261		1.549	1.344 1.338	
Owns Vasundhara/Gramakantha Paramboke land			0.690			0.883			1.405	
NLASSETS (nonland assets)										
Maximum years of education in household Number of adults (15–59)	1.075** 1.008	1.074** 1.004	1.075** 1.008	1.001 1.127	1.000 1.120	1.000 1.136	1.101*** 0.942	1.100*** 0.939	1.101*** 0.94	
Log (total value of household productive assets)	0.942**	0.941**	0.942**	1.041**	1.039**	1.039**	1.042	1.041	1.042	
Household level										
Scheduled caste	0.668	0.681	0.664	1.005	1.024	0.980	0.596	0.605	0.596	
Scheduled tribe Other backward caste	0.812 0.401**	0.819 0.406**	0.805 0.401**	2.611** 0.873	2.608** 0.885	2.511** 0.857	1.276 0.669	1.284 0.675	1.263 0.673	
Community Level										
Gajapati District	0.584	0.575	0.575	0.495***	0.486***	0.451***	0.403	0.396	0.405	
Jagatsinghpur District Khurda District	0.290*** 0.609	0.279*** 0.579	0.286*** 0.596	0.673 0.207***	0.633 0.190***	0.618 0.188***	0.590 0.716	0.575 0.686	0.586 0.715	
Baseline	10.449**	10.499**	10.480**	0.837	0.834	1.007	0.914	0.922	0.867	

To interpret these results we start by looking at the first row of Table 4.4, where we can see how likely households are to choose any of the preferred livelihood strategies. More specifically, our models predict that the average household in our sample is 10 times more likely to choose to be an off-farm wage laborer household than to be an agricultural wage laborer household. However, according to our results, the average household in our sample is just as likely to choose to be an agricultural wage laborer household as it is to be a wage laborer with a farm or a household with a farm and self-employment.

These relative probabilities can change as soon as we depart from the average household. The remaining numbers in the table tell us the multiplicative impact of each of the factors to this baseline relative probability. Because they are multiplicative factors, numbers higher (lower) than one mean that the factor increases (decreases) the likelihood that the household would choose the more food secure livelihood strategy by the column heading.

We start by examining which factors may enable or constrain a household from becoming an off-farm wage laborer household (rather than a more food insecure agricultural wage laborer household). Although the average household is 10 times more likely to choose to be an off-farm wage laborer household, our results suggest that households are less likely to adopt this livelihood strategy if they are located in Jagatsinghpur District or if they are classified as *other backward caste*. Neither access to nor ownership of land appears to affect the choice of livelihood strategy. However, households that have been settled for longer are less likely to be off-farm wage laborer households and so have more productive capital. On the other hand, increases in education appear to enable households to become off-farm wage laborer households.

A similar analysis allows us to distill factors that might explain which households are more likely to opt for being wage laborers with a farm rather than following the more food insecure livelihood strategy of agricultural wage laborer. Compared to households in Ganjam or Jagatsinghpur, households in Gajapati and Khurda are less likely to become wage laborers with a farm. This strategy is more likely to be chosen by households in STs and those with a higher value of productive assets. Land matters. Households that have access to more land and households that own land are more likely to become wage laborers with a farm. More specifically, households that have formally documented land and those that have inherited land are more likely to adopt this livelihood strategy.

Finally, those with access to more land and with higher levels of education are more likely to be a farm and self-employment household than to be an agricultural wage laborer household.

To summarize, being in disadvantaged castes may prevent households from adopting higher food secure livelihood strategies, while education, access to land, and ownership of land may act as enabling factors. Furthermore, the district in which a household resides plays a role in determining livelihood strategies.

These results, while interesting, mainly confirm what conventional wisdom and existing research would have suggested about the relationship between land and livelihood. As the development community increasingly recognizes the need to design gender-sensitive projects and programming, it is finding that the Basic Model is limited in what it can contribute. We now turn to the Gendered Model, aligned with the GAAP framework, to look for additional insights.

## Results from the Gendered Model: Assets and Livelihood Strategies

Table 4.5 shows the results of the Gendered Model, which disaggregates several of the asset-related variables by sex and allows us to see how sex-specific asset ownership influences households' choice of livelihood strategy.

Table 4.5 Gendered Model: Multinomial logit model to determine households' livelihood strategies (N = 1,578)

8					0	` /	,		
Covariates	Probability that households choose the off-farm wage laborer strategy over being an agricultural wage laborer household			probability that households choose the wage laborer with a farm strategy over being an agricultural wage laborer household					
	(1)	(2)	(3)	(1)	(2)	(3)	(1)	(2)	(3)
Baseline probability that households would choose livelihood strategy [] over being an agricultural wage laborer household	9.041*	7.902*	8.125*	0.601	0.602	0.745	0.604	0.601	0.611
LAND									
Log (land household can access)	1.105	1.109	1.119	2.181***	2.169***	2.159***	1.425***	1.420***	1.429***
Log (average number of months since land was acquired)	0.749**	0.783	0.766	0.998	1.017	0.969	0.978	0.991	0.969
Owns land * female share	1.109			1.308			2.055**		
Owns land * male share	0.880			1.273			0.935		
Owns undocumented land		0.237			0.655			0.763	
Owns undocumented land * female share		1.156			1.141			1.462	
Owns undocumented land * male share		2.539			1.646			1.387	
Owns documented land		0.148			0.981			0.808	
Owns documented land * female share		1.884			1.749			3.649**	
Owns documented land * male share		6.017			1.502			0.931	
Owns inherited land			0.247**			0.734			0.646
Owns inherited land * female share			1.738			2.165**			3.326**
Owns inherited land * male share			2.554			1.545			1.155
Owns purchased land			0.000***			0.851			0.498
Owns purchased land * female share			1.133			0.829			1.382
Owns purchased land * male share			0.000***			1.755			2.117
Owns Vasundhara/GKP land			0.476			2.513			5.377
Owns Vasundhara/GKP land * female share			0.908			0.319			0.571
Owns Vasundhara/GKP land * male share			1.662			0.669			0.395

**Table 4.5 Continued** 

Covariates	Probability that households choose the off-farm wage laborer strategy over being an agricultural wage laborer household			probability that households choose the wage laborer with a farm strategy over being an agricultural wage laborer household					
	(1)	(2)	(3)	(1)	(2)	(3)	(1)	(2)	(3)
NLASSETS (nonland assets)									
Maximum years of education among females in hh	1.001	0.996	0.997	1.027	1.024	1.027	1.042**	1.039**	1.042**
Maximum years of education among males in hh	1.095**	1.097**	1.103**	0.998	0.998	0.999	1.106***	1.106**	1.113***
Number of adult females (15–59)	0.857	0.893	0.894	0.949	0.958	0.965	0.837	0.848	0.844
Number of adult males (15–59)	1.026	1.012	1.024	1.206**	1.192	1.223**	1.068	1.056	1.052
Log (total value of household productive capital)	0.846***	0.865**	0.870**	1.047	1.047	1.051	1.076	1.079	1.083
Share of productive assets owned by female head of hh	1.714	1.752	1.652	1.660**	1.736**	1.635**	1.491	1.533	1.454
Share of productive assets owned by male head of hh	1.521	1.199	1.273	0.631	0.596	0.629	0.532	0.503**	0.538
Household level									
Scheduled caste	0.639	0.657	0.631	1.020	1.027	1.005	0.618	0.617	0.625
Scheduled tribe	0.759	0.769	0.751	2.812**	2.747**	2.759**	1.303	1.274	1.273
Other backward caste	0.351**	0.355**	0.347**	0.849	0.841	0.823	0.634	0.619	0.627
Community Level									
Gajapati District	0.596	0.574	0.580	0.488***	0.472***	0.437***	0.457	0.445	0.452
Jagatsinghpur District	0.262***	0.268***	0.282***	0.589	0.551	0.543	0.482	0.476	0.476
Khurda District	0.640	0.636	0.712	0.184***	0.165***	0.173***	0.657	0.631	0.699

Notes: hh = household; GKP = Gramakantha Paramboke. Omitted categories are Does Not Own Land, General Caste, and Ganjam District; \*\* and \*\*\* indicate significance at the 5%, and 1% levels, respectively. Standard errors were clustered by revenue circle (the next land administration level above the village level) and bootstrapped with 500 repetitions. Share of productive assets owned by female and male heads of household includes value of assets jointly owned by female and male heads.

As can be observed, all the patterns associated with factors other than assets mirror those in the Basic Model. The relative probabilities that households would choose any given strategy are similar to the ones already discussed, as is the influence of factors such as district and caste on households' choice of livelihood strategy.

Households' access to land and the amount of time that has passed since they acquired the plot are household-level variables and have not been disaggregated by sex. The extent to which they influence households' decisions and in which direction is thus similar to what the Basic Model uncovered.

For the rest of the variables, the Gendered Model can contribute to a more nuanced identification of patterns. This is what we find: An average household in our sample is between eight and nine times more likely to be an off-farm wage laborer household than an agricultural wage laborer household. However, this changes dramatically, reducing the probability that the household will be an off-farm wage laborer household, if the household has inherited land or purchased land, especially if most of the land owned by the household is owned by the male head. Similarly, while the household's education matters, with higher levels of education enabling households to become off-farm wage laborer households, what really matters is the education of the men in the household, not that of the women. This is likely to signal that wage employment opportunities (or demand) may be biased toward men and that the men who are more educated and who have not inherited or purchased land are the most likely to avail themselves of these opportunities.

Building on this discussion, the center panel of Table 4.5 sheds lights on characteristics associated with households that are more likely to choose to be wage laborers with a farm rather than agricultural wage laborer households. in line with the idea that wage opportunities may favor males, households that follow a wage laborer with a farm strategy have a significantly larger male labor force—a greater number of adult male members between the ages of 15 and 59.

Interestingly, the land and nonland productive assets owned by the households do not seem to matter, but as posited by the GAAP framework, who owns these assets does matter. When listing their households' assets, women were asked to indicate which ones they owned, either jointly with their husbands or by themselves. We find that households that are more likely to follow a wage laborer with a farm strategy are households in which women own or co-own a larger share of their households' assets. More specifically, this is the case when the woman owns a larger share of her household's productive assets and, among households that have inherited land, the woman owns a larger share of her household's land

Finally, looking at the right panel of Table 4.5, we can see that while the Basic Model revealed that education was the key factor leading households to adopt a farm and self-employment strategy, the Gendered Model indicates that the education of the men *and* the female heads of household matter. The more educated either or both of them are, the more likely their household is to follow a farm and self-employment strategy. The model also indicates that the households that are more likely to pursue this strategy are those in which women own a larger share of land, particularly among households who have documents to prove their ownership and have inherited land.

To summarize, the Gendered Model suggests that among the people we interviewed: (1) the level of education matters when households want to engage in either wage labor or self-employment, (2) wage employment appears to be heavily biased toward men, and (3) households whose livelihood strategy includes a farm are those in which women tend to own a larger share of the productive assets, particularly land.

#### 5. DISCUSSION

As the analysis in this paper demonstrates, a gendered framework and analysis allows for a more nuanced understanding of the factors—particularly assets—that enable or constrain households' ability to adopt livelihood strategies expected to yield higher food security. By exploring the relationship between households' sex-disaggregated asset portfolios and the livelihood strategies they adopt, we provide evidence-based support for a three-pronged gender-sensitive strategy to improve rural households' food security that addresses wage employment, access to education, and land rights.

First, wage labor markets in this region appear to be highly gendered. Households were significantly more likely to engage in wage work and become more food secure if they had a larger male labor force—more adult men in the household—or if the men were more educated. Despite their households' poverty and women's eligibility for the National Rural Employment Guaranty Scheme, the majority of the women in our sample described themselves as housewives. Our qualitative research suggests that in this relatively remote and hilly region of India, social norms defining what constitutes appropriate behavior for women greatly constrain their wage-earning opportunities. The wage labor available through National Rural Employment Guaranty Scheme in this area is mostly limited to earth moving for road construction, requiring hard physical labor in the company of men, an unappealing option for all but the most destitute women.

A gender-sensitive and more inclusive version of an employment guarantee scheme would effectively expand the job opportunities to encompass activities in which women can engage, provide childcare assistance, and offer safe and affordable transportation if needed. More broadly, our findings point to the need for interventions by government agencies and civil society organizations to equip women and men with the necessary skills to be employable and to work with communities to address discriminatory perceptions of what constitutes acceptable jobs for women.

Second, higher levels of education enable households to adopt livelihood strategies associated with better food security. Households with higher levels of education were significantly more likely to engage in wage labor or to be self-employed. It is important to note that the education of both men and women influences households' adoption of these more food secure livelihood strategies. This finding emphasizes the importance of investing in both boys' and girls' education, despite the common argument that girls will become *just housewives*. Moreover, given that the most food secure households have adopted livelihood strategies that include self-employment, an effective strategy to improve food security should include interventions to strengthen rural men and women's negotiation, marketing, and management skills, forms of education that extend beyond what is taught in the formal public schools and that can be integrated into youth programs and skill-development efforts.

Finally, land plays a key role. Households that are most food secure are those whose livelihood strategies include agricultural and livestock work on their farms. Households that adopt these strategies have access to more land and are more likely to own their land. They are more likely to have documents to prove their ownership and more likely to have inherited land. Importantly, results from the Gendered Model show that households that adopt these strategies tend to be those in which women own or co-own a larger proportion of their households' land. The nature of our data prevents us from rigorously determining to what extent these land-related factors enable households to adopt the more food secure livelihood strategies that include a farm or whether they are an outcome of these strategies. Yet the strong and significant associations we find are enough to prompt policy action and to support government allocation programs of the type proposed under the Homestead Act currently being discussed by the government of India.

More specifically, our results suggest that (1) the size of the plot matters, and therefore land allocation programs should resist the political and economic temptation to settle for plots that are too small; (2) having documents to prove ownership is important, and therefore land regularization programs may constitute effective tools to enhance food security; and (3) who owns the land makes a difference, making a strong case for designing land allocation and regularization programs that are gender-sensitive,

whereby staff are equipped to address gender biases in customs and processes, work in an inclusive and gender-sensitive way at all stages, and ensure that women's names are included on the land documents.

In addition, our results suggest that development interventions that seek to enable the adoption of better household livelihood strategies in this region must consider the gendered context in which they operate—men and women's employment opportunities, their skills, and the land they have—and must incorporate explicit efforts to address constraints on these three fronts.

To conclude, we should note that this analysis would have not been possible without sex-disaggregated data. This highlights the need to develop, fund, and implement systematic efforts to gather sex-disaggregated asset data at the national and state levels.

# **APPENDIX: SUPPLEMENTARY TABLES**

Table A.1 Sample composition (N = 1,730)

Districts	Blocks	Villages
	*Khallikote (156 households)	Chillipoi, Gunduripur, Marei Nuagon, Chakasingh, Badapalli, P. Balarampur
	Beguniapada (206 households)	Belapada, Burujhari Nuagon, Maridipalli, Baidipada, Chhunchipada, Jokalundi, Borabara, Nuamaridipalli, Saranianuasahi, Kalimeghi, Chakundajhola, Manipur
Ganjam (751 households)	Seragada (130 households)	Padampur, Khairbethi, Gothagaon, Kaspa, Baklikana
	^Digapahandi (119 households)	Bharamarapurpalli, Siripur, Badadumula, Ankarada, Bomkei
	^Sanakhemundi (140 households)	Konkorada, Matiaburei, Pattapur, Ambagaon, Maulabhanja
Gajapati	*Mohana (177 households)	Endrima, Dhepapadar, Kirma, Narangi, Bhaliapada, Hikirikupa, Sanakhani
(354 households)	Gumma (177 households)	Pattimul colony, Buruding, Tarava
Jagatsinghpur	*Erasama (196 households)	Nauratanpur, Padmapur, Harishpurgarh, Dhobei, Asia, Garia, Kanaguli
(374 households)	Biridi (178 households)	Chandapur, Hajipur, Kulakoijanga
Khurda (251 households)	Banapur (251 households)	Narandihi, Aranga, Dhuanali, Padanpur, Bandhamundei, Nandapur, Antarakiari, Dolamundei, Bhesari, Raipada, Kiagorada, Pari Nuagaon, Mangarajpur, Gopikartapur, Saranai, Alaidiha Patna, Kaunrapali Patana, Manitiri

Source: Authors' calculations based on household survey data.

Notes: \* denotes Landesa Vasundhara pilot area. ^ denotes GKP-eligible area.

Table A.2 Descriptive summary of variables used in the Basic Model (N = 1,695)

Variable description	Mean	Standard deviation	Minimum	Maximum
LAND				
Log (land household can access)	2.608	1.844	0	9.328
Log (average number of months since land was				
acquired)	5.294	0.794	1.386	7.343
Does not own land	0.371	0.483	0	1
Owns land	0.629	0.483	0	1
Owns land, undocumented	0.314	0.464	0	1
Owns formally documented land	0.316	0.465	0	1
Owns inherited land	0.455	0.498	0	1
Owns purchased land	0.091	0.287	0	1
Owns Vasundhara/GKP land	0.083	0.276	0	1
NLASSETS (nonland assets)				
Maximum years of education in household	6.044	3.908	0	15
Number of adults (15–59)	2.532	1.287	0	10
Total value of household productive assets (Rupees)	3016	4618	0	70000
Household Level				
Scheduled caste	0.318	0.466	0	1
Scheduled tribe	0.299	0.458	0	1
Other backward caste	0.286	0.452	0	1
General caste	0.097	0.297	0	1
Community Level				
Ganjam District	0.438	0.496	0	1
Gajapati District	0.203	0.402	0	1
Jagatsinghpur District	0.219	0.414	0	1
Khurda District	0.139	0.346	0	1

Source: Authors' calculations based on household survey data.

Note: GKP = Gramakantha Paramboke.

Table A.3 Descriptive summary of variables in the Gendered Model (N = 1,578)

Variable description	Mean	Standard. deviation	Minimum	Maximum
LAND				
	0.000	4.050	0	0.000
Log (land household can access) Log (average number of months since land was	2.668	1.853	0	9.328
acquired)	5.288	0.787	1.386	7.343
Does not own land	0.364	0.481	0	1
Owns land	0.636	0.481	0	1
Owns land, undocumented	0.314	0.464	0	1
Owns formally documented land	0.321	0.467	0	1
Owns inherited land	0.463	0.499	0	1
Owns purchased land	0.089	0.285	0	1
Owns Vasundhara/GKP land	0.084	0.277	0	1
Share of land owned by female head of household	0.380	0.463	0	1
Share of land owned by male head of household	0.526	0.472	0	1
NLASSETS (nonland assets)				
Maximum years of education among females in				
nousehold	3.939	3.917	0	15
Maximum years of education among males in	5.040	4.000	•	4.5
nousehold	5.313	4.000	0	15
Number of adult females (15–59)	1.343	0.755	0	5
Number of adult males (15–59)	1.290	0.810	0	6
Fotal value of household productive assets (Rupees) Share of productive assets owned by female head of	3118	4654	0	70000
nh	0.479	0.462	0	1
Share of productive assets owned by male head of hh	0.601	0.471	0	1
Household Level				
Scheduled caste	0.321	0.467	0	1
Scheduled tribe	0.295	0.456	0	1
Other backward caste	0.285	0.451	0	1
General caste	0.100	0.299	0	1
Community Level				
Ganjam District	0.431	0.495	0	1
Gajapati District	0.202	0.401	0	1
Jajapan District				
Jagatsinghpur District	0.224	0.417	0	1

Notes: GKP = Gramakantha Paramboke; hh = household. Share of productive assets owned by female and male heads of household includes value of assets jointly owned by female and male heads.

Table A.4 Summary of significant results using alternative estimations of basic land documentation model

	Probability that households choose the off-farm wage laborer strategy over being ar agricultural wage laborer household			Probability that households choose the wage laborer with in a farm strategy over being an agricultural wage laborer household					
Covariates	(a) MNL	(b) MNP	(c) LOGIT	(a) MNL	(b) MNP	(c) LOGIT	(a) MNL	(b) MNP	(c) LOGIT
LAND									
Log (land household can access)				+	+	+	+	+	+
Log (average number of months since land was acquired)	-	-	-						
Owns undocumented land		-	-						
Owns documented land				+	+	+			+
NLASSETS (nonland assets)									
Maximum years of education in household	+	+	+				+	+	+
Number of adults (15–59)						+			
Log (total value of household productive assets)	-	-	-	+	+				+
Household Level									
Scheduled caste									
Scheduled tribe				+	+				
Other backward caste	-	-	-						
Community Level									
Gajapati District				-	-	-			-
Jagatsinghpur District	-	-	-			-	-		-
Khurda District		-	-	-	-	-			-

Note: MNL = Multinomial logit model; LOGIT = individual logit model; MNP = Multivariate probit model. + denotes a positive, significant effect; - denotes a negative, significant effect; blank cells indicate no significant effect detected; significance determined at p<0.1.

Table A.5 Summary of significant results using alternative estimations of gendered land documentation model

		ty that hou			ty that hou			ty that hou	
	choose the off-farm wage laborer strategy over being a agricultural wage laborer household			choose the wage laborer with n a farm strategy over being an agricultural wage laborer household					
Covariates	(a) MNL	(b) MNP	(c) LOGIT	(a) MNL	(b) MNP	(c) LOGIT	(a) MNL	(b) MNP	(c) LOGIT
LAND									
Log (land household can access)				+	+	+	+	+	+
Log (average number of months since land was acquired)		-							
Owns undocumented land		-							
Owns undocumented land * female share									
Owns undocumented land * male share									
Owns documented land									
Owns documented land * female share							+	+	+
Owns documented land * male share									
NLASSETS (nonland assets)									
Maximum. years of education among females in household							+	+	
Maximum. years of education among males in household	+	+	+				+	+	+
Number of adult females (15–59)									
Number of adult males (15–59)					+	+			
Log (total value of household productive capital)	-	-							
Share of productive assets owned by female head of hh			+	+	+	+			
Share of productive assets owned by male head of hh							-		
Household Level									
Scheduled caste									
Scheduled tribe				+	+				
Other backward caste	-	-							
Community Level									
Gajapati District				-	-	-			
Jagatsinghpur District	-	-	-			-			
Khurda District				-	-	-			

Note: GKP = Gramakantha Paramboke; hh = household; MNL = Multinomial logit model; LOGIT = individual logit model; MNP = Multivariate probit model. + denotes a positive, significant effect; - denotes a negative, significant effect; blank cells indicate no significant effect detected; significance determined at p < 0.1. Standard errors were clustered by Revenue Circle (the next land administration level above the village level) and bootstrapped with 500 repetitions. Share of productive assets owned by female and male heads of household includes value of assets jointly owned by female and male heads

#### REFERENCES

- Abdulai, A., and A. Crole-Rees. 2001. "Determinants of Income Diversification Amongst Rural Households in Southern Mali." *Food Policy* 26 (4): 437–452.
- Adato, M., and R. Meinzen-Dick. 2002. Assessing the Impact of Agricultural Reserach on Poverty Using the Sustainabile Livelihoods Framework. FCND Discussion Paper 128. Washington, DC: International Food Policy Research Institute.
- Adato, M., M. Carter, and J. May. 2006. "Exploring Poverty Traps and Social Exclusion in South Aftrica Using Qualitative and Quantitative Data." *Journal of Development Studies* 42: 226–247.
- Agarwal, B. 1994. "Gender and Command over Property: A Critical Gap in Economic Analysis and Policy in South Asia." *World Development* 22 (10): 1455–1478.
- Alinovi, L., M. D'Errico, M. Erdgin, and D. Romano. 2010. "Livelihoods, Strategies and Household Resilience to Food Insecurity: An Empirical Analysis to Kenya." European Report on Development.
- Allendorf, K. 2007. "Do Women's Land Rights Promote Empowerment and Child Health in Nepal?" *World Development* 35 (11): 1975–1988.
- Ashley, C., and D. Carney. (1999a). *Sustainable Livelihoods Guidance Sheets*. London: Department for International Development.
- ——. 1999b. Sustainable Livelihoods: Lessons from Early Experience. London: Department for International Development.
- Babulo, B., B. Muys, F. Nega, E. Tollens, J. Nyssen, J. Deckers et al. 2008. "Household Livelihood Strategies and Forest Dependence in the Highlands of Tigray, Northern Ethiopia." *Agricultural Systems* 98 (2): 147–155.
- Bebbington, A. 1999. "Capitals and Capabilities: A Framework for Analyzing Peasant Viability, Rural Livelihoods, and Poverty." *World Devlopment* 27 (12): 2021–2044.
- Bekele, G., and A. Mekonnen. 2010. *Investments in Land Conservation in the Ethiopian Highlands: A Household Plot-Level Analysis of the Roles of Poverty, Tenure Security, and Market Inventives.* Washington, DC: Resources for the Future.
- Bhagowalia, P., P. Menon, A. Quisumbing, and V. Soundarrajan. 2012. What Dimensions of Women's Empowerment Matter Most for Child Nutrition? Evidence Using Nationally Representative Data from Bangladesh. IFPRI Discussion Paper 01192. Washington, DC: International Food Policy Research Institute.
- Birch-Thomsen, T., P. Frederikson, and H.-O. Sano. 2001. "A Livelihood Perspective on Natural Resource Management and Environmental Change in Semiarid Tanzania." *Economic Geography* 77 (1): 41–66.
- Brown, D., E. Stevens, J. Ouma, F. Murithi, and C. Barrett. 2006. "Livelihood Strategies in the Rural Kenyan Highlands." *The African Journal of Agricultural and Resource Economics* 1 (1): 21–36.
- Caliński, T., and J. Harabasz. 1974. "A Dendrite Method for Cluster Analysis." *Communications in Statistics—Theory and Methods* 3 (1): 1–27.
- Carter, M., and C. Barrett. 2006. "The Economics of Poverty Traps and Persistent Poverty: An Asset-Based Approach." *The Journal of Development Studies* 42: 178–199.
- Census of India. 2011. "Census Data Online." Accessed June 2013. http://censusindia.gov.in/2011-common/censusdataonline.html:.
- Chambers, R. 1995. "Poverty and Livelihoods: Whose Reality Counts?" *Environment and Urbanization* 7 (1): 173–204.
- Chambers, R., and G. Conway. 1991. Sustainable Rural Livelihoods: Pratical Concepts for the 21st Century. IDS Discussion Paper 296. Brighton, UK: Institute of Development Studies.

- Deere, C., and C. Doss. 2006. "The Gender Asset Gap: What Do We Know and Why Does It Matter?" *Feminist Economics* 12 (1–2): 1–50.
- Deere, C., and M. Leon. 2003. "The Gender Asset Gap: Land in Latin America." World Development 31 (6): 925–947.
- De Haan, L., and A. Zoomers. 2005. "Exploring the Frontier of Livelihoods Research." *Development and Change* 36 (1): 27–47.
- Deo, F. 2011. "Land Distribution Programmes in Odisha—A Social Audit in Nuapada." *Social Change* 41 (2): 271–291.
- Duflo, E. 2011. *Women's Empowerment and Economic Development*. NBER Working Paper 17702. Cambridge, MA, US: National Bureau of Economic Research.
- Ellis, F. 2000. Rural Livelihoods and Diversity in Developing Countries. Oxford, UK: Oxford University Press.
- Field, E. 2003. Fertility Responses to Land Titling: The Roles of Ownership Security and the Distribution of Household Assets. Cambridge, MA, US: Harvard University Press.
- Giovarelli, R. 2009. "Gender and Land Tenure Reform." In *One Billion Rising: Law, Land and the Alleviation of Global Poverty*, edited by R. L. Prosterman, R. Mitchell, and T. Hanstad, 195–233. Amsterdam: Leiden University Press: 195-234.
- Gupta, J. 2006. "Property Ownership of Women as Protection for Domestic Violence: The West Bengal Experience." In *Property Ownership and Inheritance Rights of Women for Social Protection—The South Asia Experience*, edited by P. Panda, J. Gupta, I. Bulankulame, N. Bhatla, S. Chakraborty, and N. Duvvury, 37–56. Washington, DC: International Center for Research on Women.
- Haddad, L., and J. Hoddinott. 1997. *Intrahousehold Resource Allocation in Developing Countries: Models, Methods and Policy*. Baltimore, MD, US: Johns Hopkins University Press.
- Iiyama, M., P. Kariuki, P. Kristjanson, S. Kaitibie, and J. Maitima. 2008. "Livelihood Diversification Strategies, Incomes, and Soil Management Strategies: A Case Study from Kerio Valley, Kenya." *Journal of International Dyelopment* 20 (3): 380–397.
- Jansen, H., J. Pender, A. Damon, W. Wielemaker, and R. Schipper. 2006. "Policies for Sustainable Development in the Hillside Areas of Honduras: A Quantitative Livelihoods Approach." *Agricultural Economics* 34 (2): 141–153.
- Katz, E., and J. Chamorro. 2002. *Gender, Land Rights, and the Household Economy in Rural Nicaragua and Honduras*. Paper prepared for USAID/BASIS CRSP, Madison, WI, US.
- Lemke, S., F. Yousefi, A. Eisermann, and A. Bellows. 2012. "Sustainable Livelihoods Approaches for Exploring Smallholder Agricultural Programs Targeted at Women: Examples from South Africa." *Journal of Agriculture, Food Systems, and Community Development* 3 (1): 25–41.
- Liang, Y., S. Li, M. Feldman, and G. Daily. 2012. "Does Household Composition Matter? The Impact of the Grain for Green Program on Rural Livelihoods in China." *Ecological Economics* 75: 152–160.
- Mearns, R., and S. Sinha. 1999. *Social Exclusion and Land Administration in Orissa, India.* Washington, DC: World Bank.
- Meinzen-Dick, R., N. Johnson, A. Quisumbing, J. Njuki, J. Behrman, D. Rubin et al. 2011. *Gender, Assets, and Agricultural Development Programs: A Conceptual Framework.* Washington, DC: International Food Policy Research Institute.
- Moser, C. 1998. "The Asset Vulnerability Framework: Reassessing Urban Poverty Reduction Strategies." World Development 26 (1): 1–19.
- NSSO (National Sample Survey Office ). 2013. *Key Indicators of Household Consumer Expenditure in India*, 2011–2012. New Delhi: Government of India.

- Nielsen, Ø., S. Rayamajhi, P. Uberhuaga, H. Meilby, and C. Smith-Hall. 2012. "Quantifying Rural Livelihood Strategies in Developing Countries Using an Activity Choice Approach." *Agricultural Economics* 44 (1): 57–71.
- Orr, A., and B. Mwale. 2001. "Adapting to Adjustment: Smallholder Livelihood Strategies in Southern Malawi." *World Development* 29 (8): 1325–1343.
- Panda, P. 2006. "Domestic Violence and Women's Ownership: Delving Deeper into the Linkages in Kerala." In *Property Ownership and Inheritance Rights of Women for Social Protection—The South Asia Experience*, edited by P. Panda, J. Gupta, I. Bulankulame, N. Bhatla, S. Chakraborty, and N. Duvvury, 10–36. Washington, DC: International Center for Research on Women.
- Pender, J., and J. Kerr. 1996. *Determinants of Farmers' Indigenous Soil and Water Conservation Investments in India's Semiarid Tropics*. Washington, DC: International Food Policy Research Institute.
- Petracco, C., and J. Pender. 2009. Evaluating the Impact of Land Tenure and Titling on Access to Credit in Uganda. Washington, DC: International Food Policy Research Institute.
- Pitt, M., O. Chowdhury, and D. Millimet. 2003. "Credit Programs for the Poor and the Health Status of Children in Rural Bangladesh." *International Economic Review* 44 (1): 87–118.
- Santos, F., V. Savath, D. Fletschner and A. Peterman. 2013. "Can Government Allocated Land Contribute to Food Security? Intrahousehold Analysis of West Bengal's Microplot Allocation Program." IFPRI Discussion Paper 01310. Washington, DC: International Food Policy Research Institute.
- Scoones, I. 1998. Sustainable Rural Livelihoods: A Framework for Analysis. Brighton, UK: Institute of Development Studies.
- ———. 2009. "Livelihoods Perspective and Rural Development." *The Journal of Peasant Studies* 36 (1): 171–196.
- Smith, L., U. Ramakrishnan, A. Ndiaye, L. Haddad, and R. Martorell. 2003. *The Importance of Women's Status for Child Nutrition in Developing Countries*. Washington, DC: International Food Policy Research Institute.
- State Government of Odisha. 2011. Annual Plan 2011–2012. Bhubaneswar, India.
- Swindale, A., and P. Bilinsky. 2006. *Household Dietary Diversity Score (HDDS) for Measurement of Household Food Access: Indicator Guide*, vol. 2. Washington, DC: Food and Nutrition Technical Assistance Project, Academy for Educational Development.
- World Bank, Food and Agriculture Organization, and International Fund for Agricultural Development. 2009. "Gender Issues in Land Policy and Administration." In *Gender in Agriculture Sourcebook*, 125–171. Washington, DC: World Bank.
- Wright, H., P. Kristjanson, and G. Bhatta. 2012. *Understanding Adaptive Capacity: Sustainable Livelihoods and Food Security in Coastal Bangladesh*. CCAFS Working Paper 32. Frederiksberg, Denmark: Climate Change, Agriculture and Food Security.

#### RECENT IFPRI DISCUSSION PAPERS

# For earlier discussion papers, please go to <a href="www.ifpri.org/pubs/pubs.htm#dp">www.ifpri.org/pubs/pubs.htm#dp</a>. All discussion papers can be downloaded free of charge.

- 1322. Exploring local perceptions of climate change impact and adaptation in rural Bangladesh. Peter Davis and Snigdha Ali, 2014.
- 1321. Closing the gender asset gap: Learning from value chain development in Africa and Asia. Agnes R. Quisumbing, Deborah Rubin, Cristina Manfre, Elizabeth Waithanji, Mara van den Bold, Deanna Olney, and Ruth Meinzen-Dick, 2014.
- 1320. The political economy of Zambia's recovery: Structural change without transformation? Danielle Resnick and James Thurlow, 2014.
- 1319. *The impact of Bolsa Família on schooling: Girls' advantage increases and older children gain*. Alan de Brauw, Daniel O. Gilligan, John Hoddinott, and Shalini Roy, 2014.
- 1318. Exploring strategic priorities for regional agricultural research and development investments in Southern Africa. Michael Johnson, Samuel Benin, Liangzhi You, Xinshen Diao, Pius Chilonda, and Adam Kennedy, 2014.
- 1317. How do intrahousehold dynamics change when assets are transferred to women?: Evidence from BRAC's Challenging the Frontiers of Poverty Reduction—Targeting the Ultra Poor Program in Bangladesh. Narayan Das, Rabeya Yasmin, Jinnat Ara, Md. Kamruzzaman, Peter Davis, Julia A. Behrman, Shalini Roy, and Agnes R. Quisumbing, 2013.
- 1316. Is there an app for that?: The impact of community knowledge workers in Uganda. Bjorn Van Campenhout, 2013.
- 1315. Can integrated agriculture-nutrition programs change gender norms on land and asset ownership: Evidence from Burkina Faso. Mara van den Bold, Abdoulaye Pedehombga, Marcellin Ouedraogo, Agnes R. Quisumbing, and Deanna Olney, 2013.
- 1314. Moving in the right direction?: Maize productivity and fertilizer use and use intensity in Ghana. Antony Chapoto and Catherine Ragasa, 2013.
- 1313. Women's empowerment in agriculture, production diversity, and nutrition: Evidence from Nepal. Hazel Jean L. Malapit, Suneetha Kadiyala, Agnes R. Quisumbing, Kenda Cunningham, and Parul Tyagi, 2013.
- 1312. Don't tell on me: Experimental evidence of asymmetric information in transnational households. Kate Ambler, 2013.
- 1311. Can dairy value-chain projects change gender norms in rural Bangladesh?: Impacts on assets, gender norms, and time use. Agnes R. Quisumbing, Shalini D. Roy, Jemimah Njuki, Kakuly Tanvin, and Elizabeth Waithanji, 2013.
- 1310. Can government-allocated land contribute to food security?: Intrahousehold analysis of West Bengal's microplot allocation program. Florence Santos, Diana Fletschner, Vivien Savath, and Amber Peterman, 2013.
- 1309. The National Health Insurance Scheme in Ghana: Implementation challenges and proposed solutions. Gissele Gajate-Garrido and Rebecca Owusua, 2013.
- 1308. Gender inequalities in ownership and control of land in Africa: Myths versus reality. Cheryl Doss, Chiara Kovarik, Amber Peterman, Agnes R. Quisumbing, Mara van den Bold, 2013.
- 1307. Farmer preferences for drought tolerance in hybrid versus inbred rice: Evidence from Bihar, India. Patrick S. Ward, David L. Ortega, David J. Spielman, Vartika Singh, 2013.
- 1306. Borrowing from the insurer: An empirical analysis of demand and impact of insurance in China. Yanyan Liu, Kevin Chen, Ruth Hill, and Chengwei Xiao, 2013.
- 1305. Organizational partnerships for food policy research impact: A review of what works. Athur Mabiso, Teunis Van Rheenen, and Jenna Ferguson, 2013.
- 1304. Fertilizer in Ethiopia: An assessment of policies, value chain, and profitability. Shahidur Rashid, Nigussie Tefera, Nicholas Minot, and Gezahengn Ayele, 2013.
- 1303. The global landscape of poverty, food insecurity, and malnutrition and implications for agricultural development strategies. Derek Headey, 2013.
- 1302. Leveling with friends: Social networks and Indian farmers' demand for agricultural custom hire services. Nicolas Magnan, David J. Spielman, Travis J. Lybbert, and Kajal Gulati, 2013.

# INTERNATIONAL FOOD POLICY **RESEARCH INSTITUTE**

www.ifpri.org

## **IFPRI HEADQUARTERS**

2033 K Street, NW Washington, DC 20006-1002 USA Tel.: +1-202-862-5600 Fax: +1-202-467-4439

Email: ifpri@cgiar.org